Author's response to reviews

Title: Dental and prosthodontic status of an over 40 year-old population in Shandong Province, China

Authors:

Qian Zhang (q.zhang@dent.umcn.nl)
Dick J Witter (d.witter@dent.umcn.nl)
Ewald M Bronkhorst (e.bronkhorst@dent.umcn.nl)
Nico HJ Creugers (n.creugers@dent.umcn.nl)

Version: 5 Date: 30 May 2011

Author's response to reviews:

Dear Editor:

Thank you for your instructions. The following adjustments were made according to your instructions:

Title: The title is now identical in the manuscript and the submission system

Abstract: The Abstract has now 348 words and is now identical in the manuscript and the submission system

Highlighting/tracking: We removed all highlighting and tracking from the manuscript.

Typography: We have checked our manuscript for any typographical errors and made final corrections.

We have checked the figures and found them to be correct.

we hope that the manuscript can now be published,

Sincerely,

Qian Zhang